Author's response to reviews

Title: Type specific Real time PCR for detection of human herpes virus 6 in schizophrenia and bipolar patients: a case control study

Authors:

Jila Yavarian (nilj57156@yahoo.com)
Somaye Shatizadeh Malekshahi (shatizadeh@razi.tums.ac.ir)
Roya Yavarian (royayavar@varian93@gmail.com)
Shaghayegh Yazdani (yazdani.vrs2@gmail.com)
Leila Janani (Leila_janani@yahoo.com)
Nazanin Zahra Shafiei Jandaghi (nz-shafiei@tums.ac.ir)
Seyed Jalal Kiani (jalalkeyani@yahoo.com)
Hamid Reza Ahmadkhaniha (dr_ahmadvaniha@yahoo.com)

Version: 6 Date: 16 September 2015

Author's response to reviews: see over
Dear Dr Joanne Ryan,

Thanks a lot for considering our article with very important criticisms and so sorry for bothering you with different revisions. I don’t know how to appreciate your patience and help in improving our manuscript. We added and improved the article as you suggested in the manuscript.

About reference number 30, as there is no published data in this regard and the findings were interesting, we added to the reference that it is unpublished data but if you recommend we will omit all the sentence and reference.

Could you please let me know if it needs some more corrections?

Your consideration greatly appreciated.

I am eagerly looking forward to hearing from you.

Yours sincerely,

Dr Jila Yavarian

On behalf of Dr H. Ahmadkhaniha